Author's response to reviews

Title: Improvement in symptoms and signs in the forefoot of patients with rheumatoid arthritis treated with anti-TNF therapy

Authors:

Catherine J Bowen (cjb5@soton.ac.uk)
Christopher J Edwards (cedwards@soton.ac.uk)
Lindsey Hooper (lhooper@soton.ac.uk)
Keith Dewbury (keith.dewbury@suht.swest.nhs.uk)
Madeleine Sampson (Madeline.Sampson@suht.swest.nhs.uk)
Sally Sawyer (sally.sawyer@suht.swest.nhs.uk)
Jane Burridge (jhb1@soton.ac.uk)
Nigel K Arden (nka@mrc.soton.ac.uk)

Version: 3 Date: 12 May 2010

Author's response to reviews: see over
Dr H Menz
Editor
The Journal of Foot and Ankle Research

12th May 2010

Dear Dr Menz,

Re: “Improvement in symptoms and signs in the forefoot of patients with Rheumatoid Arthritis treated with anti-TNF therapy”

We submitted the above manuscript for consideration of publication within ‘the Journal of Foot and Ankle Research’ as an original article and would like to take this opportunity to thank the reviewers for their continued very helpful and favourable comments. As recommended, we have now revised the flow chart (Figure 2) within our manuscript.

Yours sincerely

Dr Catherine J Bowen
Lecturer
School of Health Sciences
Direct tel: +44 (0)23 8059 7637
email: cjb5@soton.ac.uk